Author’s response to reviews

Title: Access to primary healthcare services for the Roma population in Serbia: a secondary data analysis.

Authors:

Leanne C Idzerda (lidzerda@uottawa.ca)
Orvill Adams (orvill@orvilladams.com)
Jonathan Patrick (patrick@telfer.uottawa.ca)
Ted Schrecker (tschrecker@gmail.com)
Peter Tugwell (elacasse@uottawa.ca)

Version: 4 Date: 29 June 2011

Author’s response to reviews:

Dear Editors:

Please see a list of the changes that we have made below:

Major revisions (we require the author to make these changes)

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs) - Corrected

Structure: this manuscript does not have the correct structure. Please reformat it to use these headings:

Abstract; Background; Methods; Results; Discussion; Conclusions; Abbreviations (if any); Competing interests; Authors’ contributions; Authors¿ information (optional); Acknowledgements (optional); References - Corrected

References: Please make sure references are cited in numerical order. - Double checked and corrected

Tables: the first table referenced in the manuscript text should be labelled Table 1. The second, Table 2, etc - Corrected

Box: unfortunately we cannot incorporate `box 1¿, or text boxes. Please either change the title to `Table 1¿, please remember to up-date any references to `Box 1¿ within the text, or include the information within the manuscript text- you can use indentation to highlight the text - We made the ‘boxes’ into tables.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Affiliations: Please reformat as a list of affiliations with the authors names removed and linked via superscript to the author list in numerical order – Done
Highlighting/tracking: Please remove all highlighting and tracking from the manuscript – Done

Tables- please remove the visible vertical lines from your tables – Done

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance. – Done

Thank you for your consideration,
Kind regards,
Leanne Idzerda